Author's response to reviews

Title: Use of quantification of vision by visual evoked potentials in visual disability assessment

Authors:
Jihoon Jeon Dr. (jjh786@hanmail.net)
Sungeun E Kyung Dr. (Kyungeye@gmail.com)

Version: 2 Date: 21 February 2012

Author's response to reviews:

1. Competing Interest
The author(s) declare that we have no competing interests.

2. Ethic Statement ( I changed it in method section)
The study conformed to the guidelines of the Declaration of Helsinki

3. Consent Statement ( I changed it in method section)
The written informed consent was obtained from participants.

4. Authors Contribution
JH participated in the design of the study and performed the statistical analysis.
SE conceived of the study, and participated in its design and coordination and helped to draft the manuscript. All authors read and approved the final manuscript. Each author has participated sufficiently in the work to take public responsibility for appropriate portions of the content.

5. Acknowledgement
none